## **Response to Reviewers**

Date: [Insert Date]

Editor: [Insert Editor's Name]

Journal: [Insert Journal Name]

Dear [Editor's Name],

We would like to express our gratitude to you and the reviewers for the constructive feedback on our manuscript titled "[Manuscript Title]" (Manuscript ID: [Insert Manuscript ID]). We appreciate the time and effort devoted to the review process.

## **Response to Reviewers' Comments**

Reviewer #1

**Comment 1:** [Insert Reviewer 1 Comment]

**Response:** [Insert Response]

**Comment 2:** [Insert Reviewer 1 Comment]

**Response:** [Insert Response]

Reviewer #2

**Comment 1:** [Insert Reviewer 2 Comment]

**Response:** [Insert Response]

**Comment 2:** [Insert Reviewer 2 Comment]

**Response:** [Insert Response]

## **Conclusion**

We believe that the revisions made in response to the reviewers' comments have strengthened the manuscript significantly. We hope that the changes meet the approval of you and the reviewers.

Thank you for considering our revised manuscript. We look forward to your positive response.

Sincerely,
[Your Name]
[Your Affiliation]
[Your Email]